Author’s response to reviews

Title: Effectiveness of proactive telephone counselling for smoking cessation in parents: Study protocol of a randomized controlled trial

Authors:

Kathrin Schuck (k.schuck@bsi.ru.nl)
Roy Otten (r.otten@bsi.ru.nl)
Marloes Kleinjan (m.kleinjan@bsi.ru.nl)
Jonathan B. Bricker (jbricker@fhcrc.org)
Rutger C. M. E. Engels (r.engels@bsi.ru.nl)

Version: 2 Date: 29 August 2011

Author’s response to reviews:

In response to the editorial assessment we have submitted proofs of approval of the ethical committee and external funding by e-mail.

The manuscript remains unchanged for the revision.